CERES Processing Oversight Board Meeting Notes
April 15, 2015

Meeting began at 1:00 p.m. in B2102 R263.

Meeting Notes
There were no comments or questions on the notes from the last meeting.

Announcements
e CERES Science Team Meeting will be held May 5 — 7 in the Pearl Young Theater.

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting)
TISA Gridding — Updated to document additional changes to the delivery.

PR_Database 1059 — Updated to document additional bug fixes.

TISA Averaging 1069 — Updates to Edition3 SSF1deg PGE CER10.0P4.

Inversion 1070 — Updated the SORCE TSI file to allow processing through 12/31/14.
Perl_Lib 1071 — Updates to test the new Power 7 hosts.

There were no issues identified, so these SCCRs will be approved.

CATALYST Status
Enabling SSL support in CATALYST server and PR Tool.

PR Tool Status
Planning to deliver updates for 1059 this afternoon.

Production Processing Status

Updates to posted report:

Table 2 — Edition1-CV: Terra processed through 02/18/15, Aqua processed through 12/31/14.
Table 5 — Edition3A Terra and Aqua processed through 11/14.

Table 7 — Beta2-Edition4 Terra processed through 01/07/11.

Table 9 — Edition4A: Terra processed through 06/23/08, Aqua processed through 05/31/08.

System Status
Network outage scheduled for April 25.

Status of Open SCCRs

1059 — Planning to deliver updates today.

1010 — Plan to run ValRs later this week. Should promote to PPE today.

1023 — Dennis approved the PR on 4/13/15.

1047 — Continuing operational testing.

1064 — Released 4/13/15.

1038 — New target delivery date is 4/17/15. Jonathan would like this to be in production by May
1.

1024 — Need PRs.

1017 — SIT running test cases from the test plan.

1025 — Need PR.

1058 - No updates.
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1049 — No updates.
1050 — No updates.
1052 — No updates.
1060 — No updates.
1061 — No updates.
1065 — No updates.
1066 — No updates.
1067 — No updates.
1068 — No updates.
1070 — Released 4/14/15. Needs to be promoted quickly.
1071 — No updates.

Meeting ended at 1:43 p. m.
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